**Author’s response to reviews**

**Title:** Screw migration and oesophageal perforation after surgery for osteosarcoma of the cervical spine: A case report.

**Authors:**

Luca Denaro (luca.denaro@yahoo.it)

Umile Giuseppe Longo (g.longo@unicampus.it)

Alberto Corrado Di Martino (ug.longo@unicampus.it)

Nicola Maffulli (maffulli@unicampus.it)

Vincenzo Denaro (denarcampus@gmail.com)

**Version:** 2 **Date:** 08 Dec 2017

**Author’s response to reviews:**

Many thanks for the opportunity to review this paper.

1) **Authors contributions:** the new author added to the paper (Alberto Corrado Di Martino) does not appear in the authors contributions lists. Please outline his contribution.

Answer: We state "UGL, ACDM and LD made substantive intellectual contributions to the published study and wrote the paper, evaluated the clinical conditions and the clinical course of the patient during the follow-up period". We hope this is acceptable

2) **Availability of data and materials:** it is not appropriate to state 'Not applicable'.

Answer: Data are contained within the manuscript